Author’s response to reviews

Title: Strategy for recruitment and factors associated with motivation and satisfaction in an interventional study with 210 healthy volunteers without financial compensation

Authors:

Quentin Luzurier (quentin.luzurier@chu-rouen.fr)
Cedric Damm (cedric.damm@chu-rouen.fr)
Fabien Lion (fabien.lion@gustaveroussy.fr)
Carine Daniel (carine.daniel@chu-rouen.fr)
Lucille Pellerin (lucille.pellerin-kurtz@chu-rouen.fr)
Marie Pierre Tavolacci (marie-pierre.tavolacci@chu-rouen.fr)

Version: 4 Date: 26 August 2014

Author’s response to reviews:

Re: Revision of the manuscript reference No. 2397861481264402 - Feasibility, recruitment and factors associated with motivation and satisfaction in an interventional study with 210 healthy volunteers without financial compensation

Dear Sir or Madam,

We would like to thank you and both Reviewers for providing us with constructive feedback that improves our manuscript. We have now made the requested revisions manuscript, and respond to each of the Reviewers’ comments point-by-point.

Please find, herewith, the online copy of the revised manuscript “Strategy of recruitment and factors associated with motivation and satisfaction in an interventional study with 210 healthy volunteers without financial compensation” (modified title) and responses to Reviewers’ comments.

We would like to resubmit this revised version for publication in BMC Medical Research Methodology. We hope that the revised manuscript is now acceptable for publication.

All the authors approved the revised manuscript and declare no potential competing interests. The manuscript has not been published previously, and is not under consideration for publication elsewhere.

The corresponding author is:
Marie Pierre Tavolacci, MD, MPH
Rouen University Hospital, Clinical Investigation Center CIC 1404,
1 Rue de Germont, Rouen Cedex 76031, France
Thank you for your patience and interest.
Yours sincerely,

Quentin Luzurier, Public Health Resident
Clinical Investigation Centre - INSERM 1404
Rouen University Hospital, France